Author's response to reviews

Title: An analysis of trends and determinants of health insurance and healthcare utilisation in the Russian population between 2000 and 2004: the 'inverse care law' in action

Authors:

Francesca Perlman (francesca.perlman@lshm.ac.uk)
Dina Balabanova (dina.balabanova@lshm.ac.uk)
Martin McKee (martin.mckee@lshm.ac.uk)

Version: 6 Date: 23 April 2009

Author's response to reviews:

I have made some minor editing changes to increase clarity. These are mostly to the abstract (its introductory section), the background section and a couple of sentences of the discussion of findings. (No changes were made to the methods or results sections or the tables)